Author's response to reviews

Title: Utility of Procalcitonin (PCT) and Mid regional pro-Adrenomedullin (MR-proADM) in risk stratification of critically ill febrile patients in Emergency Department (ED). A comparison with APACHE II score.

Authors:

Francesco Travaglino (francesco.travaglino@gmail.com)
Benedetta De Berardinis (benedetta.db@gmail.com)
Laura Magrini (magrinilau@libero.it)
Cristina Bongiovanni (c3bongio@yahoo.it)
Marcello Candelli (mcandelli@gmail.com)
Nicolò Gentiloni Silveri (n.gentiloni@rm.unicatt.it)
Jacopo Legramante (legramante@med.uniroma2.it)
Alberto Galante (galante@uniroma2.it)
Patrizia Cardelli (patrizia.cardelli@uniroma1.it)
Gerardo Salerno (sagerardo@libero.it)
Salvatore Di Somma (salvatore.disomma@uniroma1.it)

Version: 6 Date: 3 July 2012

Author's response to reviews:

Dear Editor,

We checked the references cited throughout the manuscript “Utility of Procalcitonin (PCT) and Mid regional pro-Adrenomedullin (MR-proADM) in risk stratification of critically ill febrile patients in Emergency Department (ED). A comparison with APACHE II score”. The revisions are the following:

1. page 3: reference n. 3 has been moved from line 11 to line 10
2. page 3: reference n.4 and n. 5 have been inverted and reference n. 4 has been moved to line 16
3. page 3: the statement at line 17 has been corrected, as suggested by the Associate Editor
4. page 9: references at line 3, line 19, line 21, line 22 and line 24 have been rearranged
5. page 10: references at line 11 have been rearranged

Thank you.

Best regards,

Francesco Travaglino, MD
Emergency Medicine Department
Sant'Andrea Hospital
“La Sapienza” University of Rome – School of Medicine and Psychology